Author's response to reviews

Title: Six weeks of home enteral nutrition versus standard care after oesophagectomy or total gastrectomy for cancer: study protocol for a randomised controlled trial

Authors:

- David J Bowrey (djb57@le.ac.uk)
- Melanie Baker (melanie.baker@uhl-tr.nhs.uk)
- Vanessa Halliday (vanessa.halliday@sheffield.ac.uk)
- Anne L Thomas (at107@le.ac.uk)
- Ruth Pulikottil-Jacob (r.jacob@warwick.ac.uk)
- Karen L Smith (kls27@le.ac.uk)

Version: 4 Date: 29 April 2014

Author's response to reviews:

Dear Ms McGavin

Please find an amended copy of the manuscript

Best wishes

David Bowrey